Author's response to reviews

Title: Pancreatic cancer-related cachexia: influence on metabolism and correlation to weight loss and pulmonary function

Authors:

Jeannine Bachmann (Bachmann@chir.med.tu-muenchen.de)
Knut Ketterer (Ketterer@chir.med.tu-muenchen.de)
Christiane Marsch (Christiane.marsch@kochstefan.de)
Kerstin Fechtner (Kerstin.fechtner@med.uni-heidelberg.de)
Holger Krakowski-Roosen (H.Krakowski-Roosen@dkfz.de)
Markus W Buechler (Markus.buechler@med.uni-heidelberg.de)
Helmut Friess (Friess@chir.med.tu-muenchen.de)
Marc E Martignoni (Martignoni@chir.med.tu-muenchen.de)

Version: 8 Date: 13 July 2009

Author's response to reviews:

Response to the Editor’s Comments:

Thank you for accepting our manuscript “Pancreatic cancer-related cachexia: influence on metabolism and correlation to weight loss and pulmonary function”.

Please find the actual version of the manuscript including the required changes.

We have added the information, that all authors have read and approved the manuscript, in the author’s contributions part.

We included the email address of the authors on the title page.

We have deleted the phrase 'Bachmann et al' from the top of each page.

Finally we have again checked the manuscript for spelling mistakes.

Yours sincerely,

Jeannine Bachmann, MD
Marc Martignoni, MD